Author's response to reviews

Title: Gastrointestinal stromal tumor with nephrotic syndrome as a paraneoplastic syndrome: Report of a case

Authors:

Kiyoko Takane (kiyoko1205@hotmail.com)
Yutaka Midorikawa (mido-tsky@umin.ac.jp)
Shintaro Yamazaki (yamazaki-nmed@umin.ac.jp)
Takahiro Kajiwara (kajikaji714@yahoo.co.jp)
Naoki Yoshida (keep-on-the-hard-hearts@dokomo.ne.jp)
Yoshiaki Kusumi (kusumi.yoshiaki@nihon-u.ac.jp)
Tadatoshi Takayama (takayama.tadatoshi@nihon-u.ac.jp)

Version: 2
Date: 26 October 2013

Author's response to reviews: see over
Dear Professor Michael Kidd,

Enclosed please find the manuscript entitled as “Gastrointestinal stromal tumor with nephrotic syndrome as a paraneoplastic syndrome: Report of a case” by Takane K et al. which we would like to be considered for publication in *Journal of Medical Case Report*. This paper is a rare case which nephrotic syndrome has improved by a gastrointestinal stromal tumor removal. Consequently, the paper provides new insights of a gastrointestinal stromal tumor.

We appreciate your review of this work. Neither the submitted paper nor any similar paper, in whole or in part, other than an abstract or preliminary communication, has been or will be submitted to or published in any other primary scientific journal. All of the authors are aware of and agree to the content of the paper and to being listed as an author. Authorization has been given to use any information conveyed by either personal communication or release of unpublished experimental data.

All correspondence should be sent to Yutaka Midorikawa at, Department of Digestive Surgery, Nihon University School of Medicine, 30-1 Oyaguchikami-machi, Itabashi-ku, Tokyo 173-8610, Japan.
Phone: +81-3-3972-8111; Fax: +81-3-3957-8299.

Sincerely yours,

Yutaka Midorikawa, MD
Nihon University School of Medicine